Author’s response to reviews

Title: Antimicrobial, antimycobacterial and antibiofilm properties of Couroupita guianensis Aubl. fruit extract

Authors:

Naif Abdullah Al-Harbi Dr (naalharbi@ksu.edu.sa)
Chandrasekar Balachandran (balaeri09@gmail.com)
Michael Karunai raj (Karunairajm@orchidpharma.com)
Veeramuthu Duraipandiyan (avdpandiyan@yahoo.co.in)
Chinnasamy Muthukumar (cmkumar1979@gmail.com)
Savarimuthu Ignacimuthu (entolc@yahoo.com)
Inshad Ali Khan (iakhan@iiim.ac.in)

Version: 2 Date: 15 October 2012

Author’s response to reviews: see over
Dear Editor

Wishes and greetings!

Thank you for sending our article for revision. Herewith we are sending the revised manuscript along with list of corrections carried out as suggested by the reviewers. The corrections have been indicated in bold letter. Kindly consider our manuscript for publications.

Thank you

With regards

Dr. Naif Abdullah-Al-Dhabi